Author's response to reviews

Title: Perception of epidemic's related anxiety in the General French Population
A Cross-sectional Study in the Rhone-Alpes Region

Authors:

Mitra Saadatian-Elahi (mitra.elahi@hu-lyonh.fr)
Françoise Facy (francoise.facy@vesinet.inserm.fr)
Corrinne Del Signore (corinne.del-signore@adm.univ-lyon1.fr)
Philippe Vanhems (philippe.vanhems@chu-lyon.fr)

Version: 4 Date: 18 March 2010

Author's response to reviews: see over
Dear Reviewer;

Below; please find a summary of the modifications which we made in the manuscript to answer your requests:

**Reviewer: Arja R R Aro**

- As requested, the term “Risk perception” was replaced by “anxiety” or “anxiety generated by epidemic risk” throughout the manuscript.
- The word “only” was removed from the abstract, result section, line 4.
- The word “educational” was added between “no” and “qualification” in the result section, lines 2-3.

Yours sincerely

Mitra Saadatian-Elahi